Author’s response to reviews

Title: Paediatric vision screening by non-healthcare volunteers: Evidence based practices

Authors

Kourosh Sabri (sabrik@mcmaster.ca)
Bethany Easterbrook (easterb@mcmaster.ca)
Nisha Khosla (nishakhosla.nk@gmail.com)
Clint Davis (DavisC@hwcdsb.ca)
Forough Farrokhyar (farrokh@mcmaster.ca)

Version: 5 Date: 14 Feb 2019

Author’s response to reviews:

1. Please change the heading labelled 'Introduction' to read 'background'
   • This heading has been changed as requested and the change can be found on Page 3, Line 1.
2. Please change the heading labelled 'Materials and Methods' to read 'Methods'
   • This heading has been changed as requested and the change can be found on Page 4, Line 10.
3. Please state all author names as initials
   • All instances of author names have been changed to include first initial and full last name. These can be found on Page 1, lines 2-3 and Page 15, lines 14-18
4. Please ensure all references are formatted in the BMC Series style
   • References have been changed to reflect BMC Medical Education style guidelines. This can be seen Pages 16-18 on all lines.